Author's response to reviews

Title: Concordance between muscle mass assessed by bioelectrical impedance analysis and by dual energy X-ray absorptiometry: a cross-sectional study

Authors:

Fanny Buckinx (fanny.buckinx@ulg.ac.be)
Jean-Yves Reginster (jyreginster@ulg.ac.be)
Nadia Dardenne (ndardenne@ulg.ac.be)
Jean-Louis Croisier (jlcroisier@ulg.ac.be)
Jean-François Kaux (jfkaux@chu.ulg.ac.be)
Charlotte Beaudart (c.beaudart@ulg.ac.be)
Justine Slomian (jislomian@ulg.ac.be)
Olivier Bruyère (olivier.bruyere@ulg.ac.be)

Version: 5 Date: 19 February 2015

Author's response to reviews: see over
Author’s response

MS: 1360472572139615
Research article
Concordance between muscle mass assessed by bioelectrical impedance analysis and by dual energy X-ray absorptiometry: a cross-sectional study Fanny Buckinx, Jean-Yves Reginster, Nadia Dardenne, Jean-Louis Croisier, Jean-François Kaux, Charlotte Beaudart, Justine Slomian and Olivier Bruyère BMC Musculoskeletal Disorders (Section: Epidemiology of musculoskeletal disorders)

Dear Miss Buckinx,

Your manuscript has now been assessed by one of our Editor and her comments are as follows:

1. Line 262
Please remove the duplicated text "is greatly reduced" from line 262.
Author: The duplicated text was deleted.

2. Kindly add a List of abbreviations section to your manuscript. Please note that Research article require the following sections:
-Background
-Methods
-Results
-Discussion (Results and Discussion may be combined) -Conclusions -List of abbreviations used (if any) Please ensure that these sections are present and clearly labelled as described above. Please do check the instructions for authors on the journal website to ensure that your manuscript follows the correct structure for this journal and article type, and to ensure that you are aware of additional recommendations for formatting that will facilitate handling of your manuscript.
Author: A list of abbreviations was added to the manuscript, after the conclusions.

List of abbreviations
BIA: Bioelectrical Impedance Analysis; DXA: Dual Energy X-ray Absorptiometry; ALM/ht²: Appendicular Lean Mass divided by height squared; ICC: IntraClass Coefficient; BMI: Body Mass Index.

3. By way of a section ?Acknowledgements?, please acknowledge anyone who contributed towards the article by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include the source(s) of funding for each author, and for the manuscript preparation. Authors must describe the role of the funding body, if any, in design, in the collection, analysis, and interpretation of data; in the writing of the manuscript; and in the decision to submit the manuscript for publication. Please also acknowledge anyone who contributed materials essential for the study. If a language editor has made significant revision of the manuscript, we recommend that you acknowledge the editor by name, where possible.
The role of a scientific (medical) writer must be included in the acknowledgements section, including their source(s) of funding. We suggest wording such as 'We thank Jane Doe who provided medical writing services on behalf of XYZ Pharmaceuticals Ltd.'
Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements section.
Author: The acknowledgements section was added to the manuscript.

Acknowledgements:

We thank L.M. Meijers society for lending us the InBodyS10 device.

4. You have uploaded the tables as figure files. Please remove them from the submission system and include the tables within the text file of the manuscript after the references. The tables should be formatted using the Table tool in your word processor. Please also move the table title to above the table and the legend to below the table, within the text.

Author: Tables have been added in the document after the references.

5. Please reformat your figures to conform with the journal style. Kindly use this link for more information: [http://www.biomedcentral.com/authors/figures](http://www.biomedcentral.com/authors/figures)

*Kindly remove the figure legend from the figure and include at the end of the manuscript.

Author: The figure was reformatted to conform with the journal style.